Author’s response to reviews

Title: Education level and physical functional limitations among Japanese community residents- gender difference in prognosis from stroke

Authors:

Kaori Honjo (khonjiyo@pbhel.med.osaka-u.ac.jp)
Hiroyasu Iso (iso@pbhel.med.osaka-u.ac.jp)
Ai Ikeda (ai-ikeda@umin.net)
Manami Inoue (mnminoue@ncc.go.jp)
Shoichiro Tsugane (stsugane@ncc.go.jp)

Version: 4 Date: 1 April 2009

Author’s response to reviews:

April, 1 2009
BMC Public Health
Editor-in-Chief:
Dr. Melissa Norton

MS: 1382316614227803
Education level and physical functional limitations among Japanese community residents- gender difference in prognosis from stroke

Dear Dr. Melissa Norton

Thank you for the opportunity to resubmit our manuscript. We have revised our paper following the helpful suggestions made by the reviewer.

With these modifications, we hope our paper is now acceptable for publication. We thank the editor and the reviewers again for their helpful suggestions, and look forward to hearing from you.

Sincerely,

Hiroyasu Iso, MD

Reviewer 1

Minor comment
1. page 6-paragraph 2, the reviewer suggested us to correct typos in the text. According to the reviewer’s suggestions, we corrected typos from hypothesis to hypotheses.
2. Page 11—paragraph 3, the reviewer asked us if there is something missing from the first sentence.

There is no missing in the sentence, but we changed the sentence to “The results of further stratification by gender suggested gender differences in the prognosis of stroke between men and women.”